Author's response to reviews

Title: Neovascular Glaucoma: A Retrospective Review From a Tertiary Center in China

Authors:

Na Liao (nana_jane@163.com)
Chao Hong Li (aoeight@hotmail.com)
Yuan Bo Liang (yuanboliang@gmail.com)
Hui Lv Jiang (hljiang314@163.com)
Ai Wu Fang (953736995@qq.com)
Sheng Jie Zhou (425760739@qq.com)
Ravi Thomas (ravi.thomas@qei.org.au)
Qin Mei Wang (wqm3@mail.eye.ac.cn)

Version: 4
Date: 9 November 2015

Author's response to reviews: see over
Dear editors:

Thanks for your work and suggestions. The point to point responds to the editorial requirements are listed below.

Best wishes!
Dr. Wang

Responds to Editorial Requirements:

1. Abstract
Please describe the context of your study in the beginning of subsection 'Background' of the abstract. Please complete the sentence that is currently there (for example by adding "The purpose of this study is" in the beginning). Please remember to also update the abstract that is uploaded separately in the submission system.

Abstract “Background” has been changed to “The purpose of this study is to report the prevalence, etiology, treatment and outcomes of neovascular glaucoma (NVG) in a tertiary care ophthalmic center in China.”

2. Authors’ Contributions
Please clarify the contributions of author HLJ in this section.

HLJ involved in data collection.
As for the part of authors’ list, there has been removed Yuanbo Liang and Ravi Thomas as their requirements.

3. Figure
Please note that the label of the horizontal axis of the figure is in Chinese.
Please explain in the figure legend what is meant with the label "censored" for one of the curves, and please move the figure legend to the end of the manuscript document.

Figure legends has been changed to:
“Fig.1. Kaplan-Meier survival curve of treatment outcomes in all 138 NVG patients (149 eyes). Censored data related to the ones who had not observed failure at that time (including missing data).”
It has been moved to the end of the manuscript document and the label of the horizontal axis is “Year”.

4. Tables
Please include the tables in the end of the manuscript document, instead of uploading them as an additional file.

The tables and manuscript have been put together below the references.

5. Copyedit:
We recommend that you copyedit the paper to improve the style of written English.

We are grateful for Ravi Thomas’s help who revised this manuscript document. I hope the style of written English has met the requirement now.